RESEARCH

Reproductive Health

Open Access

Contraceptive discontinuation: frequency and associated factors among undergraduate women in Brazil



Christiane Borges do Nascimento Chofakian^{1*}, Caroline Moreau^{2,3}, Ana Luiza Vilela Borges⁴ and Osmara Alves dos Santos⁴

Abstract

Background: In Brazil, high contraceptive prevalence rates coexist with high rates of unintended pregnancies. Contraceptive discontinuation may explain this context, but few studies have focused on highly educated young women in countries with low unmet need for modern contraception. This paper explores frequency and associated factors of contraceptive discontinuation among undergraduate students in Brazil within 12-months.

Methods: This retrospective cohort study was conducted among a probability sample of 1679 undergraduates of São Paulo University. Data were collected online using a contraceptive calendar. We examined factors related to monthly discontinuation of oral pills and male condoms using Generalized Estimating Equation models.

Results: Altogether, 19% of oral pill users and 48% of male condom users discontinued their method for method-related reasons within 12-months, and 18% of oral pill users and 15% of male condom users abandoned/or switched to less effective methods. Women in casual relationships were at increased odds of oral pill (OR = 1.4 [1.1-1.8]) and male condom discontinuation (OR = 1.3 [1.0-1.7]), and at increased odds of switching from oral pill to less effective or no method (OR = 1.4 [1.1-1.7]). Other associated factors were method specific. Women from lower socioeconomic status or who had multiple lifetime partners were more likely to discontinue or abandon the oral pill, while more sexually experienced women were less likely to discontinue the male condom.

Conclusion: Frequent method discontinuation in Brazil calls for greater attention to the difficulties women face when using short acting methods. Discontinuation was associated with type of partner and sexual experience highlighting the changing contraceptive needs of women at the early stages of their professional careers.

Keywords: Contraception, Contraceptive discontinuation, College students, Sexual and reproductive health, Brazil

* Correspondence: chris@usp.br

¹Direct-Entry Midwifery Program, School of Arts, Sciences and Humanities of the University of São Paulo, Av. Arlindo Bettio, 1000 – Ermelino Matarazzo, São Paulo, São Paulo 03828-000, Brazil

Full list of author information is available at the end of the article



© The Author(s). 2019 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

Resumo

Introdução: No Brasil, as altas taxas de prevalência de métodos contraceptivos coexistem com as altas taxas de gravidezes não desejadas. A descontinuidade dos métodos contraceptivos pode explicar esse contexto; entretanto, poucos estudos focam mulheres jovens com alto nível educacional, em países que apresentam baixo número de mulheres com necessidade insatisfeita por métodos contraceptivos modernos. Desse modo, o presente estudo explora a frequência e os fatores associados das descontinuidades contraceptivas entre estudantes universitárias no Brasil, em um período de 12 meses.

Métodos: Estudo de coorte retrospectivo realizado com uma amostra probabilística de 1679 graduandas da Universidade de São Paulo, Brasil. Os dados foram coletados online usando um calendário contraceptivo. Para examinar os fatores relacionados à descontinuidade mensal das pílulas orais e dos preservativos masculinos foram utilizados os modelos de Equações de Estimação Generalizadas.

Resultados: No total, 19% das usuárias da pílula oral e 48% das usuárias do preservativo masculino descontinuaram o método por razões relacionadas ao método em um período de 12 meses, e 18% das usuárias da pílula oral e 15% das usuárias do preservativo masculino abandonaram ou mudaram para um método menos eficaz. As mulheres que tinham relacionamentos casuais tiveram mais chance de descontinuar a pílula oral (OR = 1,4 [1, 8]) e o preservativo masculino (OR = 1,3 [1,0-1,7]), e tiveram mais chance de mudar a pílula oral para um método menos eficaz ou para nenhum método (OR = 1,4 [1, 7]). Outros determinantes foram específicos do tipo do método contraceptivo. As mulheres de nível socioeconômico mais baixo ou que tiveram múltiplos parceiros sexuais durante a vida tiveram uma maior probabilidade de descontinuar ou abandonar a pílula oral; todavia, as mulheres que tinham mais experiências sexuais foram menos propensas a descontinuar o preservativo masculino.

Conclusão: A frequente descontinuidade contraceptiva no Brasil chama atenção para as dificuldades que as mulheres enfrentam quando usam métodos de curta duração. A descontinuidade contraceptiva foi associada ao tipo de parceiro e à experiência sexual, destacando a necessidade de mudanças contraceptivas entre mulheres que estão nos estágios iniciais de suas carreiras profissionais.

Palavras-chave: Contracepção; Descontinuidade Contraceptiva; Universitárias; Saúde Sexual e Reprodutiva; Brasil.

Plain English summary

Contraceptive discontinuation is defined as switching a method of contraception or stopping a method altogether. In this study, a probabilistic sample of undergraduate women from University of São Paulo/Brazil answered a contraceptive calendar via an online survey. Respondents were asked about the main contraception that they used in the past 12-months. For each month in the calendar, the student provided information about the reasons for discontinuation, occurrence of pregnancy and their relational status with a partner. Of the 1679 respondents: 19% of oral pill users and 48% of condom users discontinued their method for method-related reasons within 12months, and 18% of pill users and 15% of condom users abandoned contraceptive use or switched to a less effective method (e.g., from oral contraceptives to male condom or from male condom to withdrawal). Changes in patterns of contraceptive use differed by characteristics of the respondents. Women of lower socioeconomic status and those who had had more than one sexual partner in their lifetime were more likely to stop using the pill, while more sexually experienced women (those for whom more than 1 year had passed since first sexual intercourse) were less likely to discontinue condom use. The major finding of the research was that the type of relationship, socioeconomic status, sexual experience, sexual partners, and type of contraceptive were strongly associated with contraceptive discontinuation among undergraduate students in Brazil.

Background

Brazil has recently undergone a rapid fertility transition with total fertility rates dropping below replacement, from 6.3 in 1986 to 1.8 in 2006, and parallel increase in modern contraceptive prevalence rates from 57% in 1986 to 77% in 2006 [1, 2]. Despite dramatic advances in contraceptive coverage and low unmet need for modern contraception (6%), 55.4% of pregnancies were reported as unintended in Brazil [3], a situation that is comparable to other countries with high contraceptive coverage [4–6].

In countries with moderate to high contraceptive prevalence, the majority of unintended pregnancies occur as a result of contraceptive discontinuation (defined as switching a method of contraception, stopping a method altogether, or failures due to inconsistent use) [7–9]. Considering all methods, 27.8% of all methods are discontinued for method-related reasons within 12-months in Brazil; more specifically, 21.6% of women switching their method and 10.3% abandon the method within the first year of use [10]. In this scenario, it is important to highlighted that although switching can heighten the risk of an unintended pregnancy, since failures are higher at the start of use of a method when women are still familiarizing themselves with the regimen, the risk of an unintended pregnancy are higher when a woman abandon the method or switch to less effective method [11].

Contraceptive discontinuation is associated with many factors, including external factors (such as quality of family planning services, guidelines, among others), but also individual and partner factors (such as demographic characteristics, partner dissatisfaction, fertility motivations, among others) [12, 13]. A study carried out with Brazilian women observed that method switching was more common among married users than their unmarried counterparts, and abandonment was positively associated with the parity, and inversely associated with educational level [10].

Contraceptive discontinuation is an important contributing factor to unintended pregnant and induced abortion. In the Brazilian context, where induced abortion is highly restricted, and thus often practiced in clandestine and unsafe contexts, this means that contraceptive discontinuation also contributes to maternal mortality and morbidity. Understanding contraceptive discontinuation is significant because it has negatively and positively reproductive consequences for women throughout their life course, especially among young women who the contraceptive trajectories are dynamic [14]. In addition, from the life course perspective, comprehending contraceptive and reproductive behaviors among young people are important, because family (and union) formation has experienced a major transformation over the last decade with delays in marriage [15, 16]. It means that a growing population of youth exposed to a potential risk of unintended pregnancy.

The latest data on discontinuation which mention youth in Brazil, dates back to the 1996 Demographic and Health Survey (DHS) [1] with little insights on how overall practices and user dynamics have evolved ever since. In Brazil, the method mix of contraceptives (especially long-acting reversible contraceptive methods (LARCs)) is limited. Consequently, among young women in Brazil, pills and male condoms dominate the contraceptive landscape, but they are usually associated with high discontinuation rates and moderate effectiveness [1, 8, 11, 17, 18]. Undergraduate students are especially motivated to delay childbearing until reaching professional stability [19, 20]. While they have lower pregnancy rates compared to less educated women [1, 6, 21], they are also more likely to report their pregnancies as unintended (60%), most of which are due to contraceptive discontinuation [22, 23]. In addition, highly educated Brazilian young women only represent 12.6% of women of reproductive age and therefore [16], their specific sexual and reproductive behaviors and needs are not well captured in a population based study including all women of reproductive age, such as the DHS. Therefore, the purpose of this study was to explore the frequency and the factors associated to contraceptive discontinuation for method-related reasons, and abandonment or switching to less effective method among undergraduate women.

Methods

We conducted a 12-months retrospective cohort study among a sample of undergraduate women from a public university in Brazil. São Paulo University is the largest institution of higher education and research in the country and best ranked in Latin America. The study obtained approval from the Human Research Ethics Committee of University of São Paulo School of Nursing.

Women aged 18-24 enrolled in regular undergraduate courses were eligible. They were selected by simple random sampling without replacement, based on the list of all the email addresses of students provided by the university (n = 18,193 emails from different female students). The sample included 1026 women, however considering the low response rate (10%) in the pre-test (n = 50), the sample was tripled. Thus, an online questionnaire was sent to 3078 women selected from the email list. A total of 2182 (71%) responded to the email, from which 50 refused to participate; and 453 were deemed non-eligible due to age (< 18 (n = 2)), sexual activity (n = 358 had never had sexual intercourse), and contraceptive use in the previous 12-months (n = 93 had never used contraception in the last 12-months). Our final population included 1679 female students aged 18-24 who reported they had ever had sexual intercourse and who had used contraception in the past 12-months.

Data were collected online through Google Form, using a self-administered questionnaire; which took, on average, 5-7 min to complete. The sociodemographic characteristics included: age (18-19/20-24); race/ethnicity (white/ black/others); religious affiliation (no religion/catholic/ Kardecist Spiritualism/Evangelical/others); relationship status (steady/casual); socioeconomic status (A/B: high level of income, and C-D/E: middle or low level of income) [24]; field of study (Human/Health/Exact Sciences); type of student enrollment (full-time/part-time); and campus site (São Paulo/out of São Paulo). Sexual reproductive information included: time since first sexual intercourse $(\leq 1 \text{ year}/2-3 \text{ years}) \geq 4 \text{ years});$ number of sexual partners (1 partner/2–3 partners/ \geq 4 partners); and previous pregnancy (no/yes). Women also completed a contraceptive calendar on a month-by-month basis covering the previous 12-months. The calendar was derived from the DHS survey [9], and collected monthly information on partnership status, contraceptive method used, and reasons for discontinuation if women reported different methods two consecutive months. Users of coital dependent methods were asked to indicate if they still considered themselves as using their method even in months when they had no sexual activity in order not to conflate coital method discontinuation with the absence of sexually activity [25].

Therefore, in this context, we defined two measures of discontinuation, which represent different levels of risk relative to unintended pregnancy. The first measure, "discontinuation for method-related reasons," classifies all women who discontinue their method but are still in need of contraception (with a partner, not pregnant, not trying to conceive, not sterile) as having discontinued for method-related reasons. The second, "abandoned or switched to less effective method", includes women who switch to a less effective method within a month or stop their method altogether while still in need of contraception [26]. Switching to a less effective method was defined as either switching from oral contraceptives to barrier methods, spermicide, fertility awareness, or withdrawal; or switching from male condom to withdrawal, fertility-awareness, or spermicide [17]. We combined abandonment and decreased efficacy in a single category due to the very small number of women who reported abandoning their method. While pregnancy risk is higher for non-users as compared to users of other less effective methods, we considered that the drop in contraceptive efficacy was an important indicator of contraceptive use dynamics [17, 27, 28].

Statistical analysis

We first described patterns of contraceptive use according to method type (pill, male condom) at the start of the contraceptive calendar (12-months prior to the survey, which we consider to be baseline). We focused on pill and condoms because together they represent 95% of the method mix in this population; thus, we did not consider other methods, such as withdrawal/fertilityawareness/spermicide/diaphragms (n = 91 women). We then turned to method specific monthly-level analysis of pill and condom discontinuation, examined across 12, 378 months of pill use and 3581 months of male condom use. We explored bivariate associations between women's sociodemographic characteristics, their sexual and reproductive health characteristics and their monthly probability of discontinuing the pill or the condom, according to our two indicators of discontinuation. In this analysis, partnership type was a monthly time varying measure. We then conducted multivariate analysis using a Generalized Estimating Equation (GEE) to investigate the odds of monthly discontinuation of the pill and condom (using both discontinuation indicators), while accounting for intra-correlation among months belonging to the same woman. We used an exchangeable correlation structure with logit link transformation function. The selection of the most appropriate correlation structure was based on the Quasi-likelihood under the Independence Model Criterion (QIC). Adjusted odds ratios and 95% confidence intervals were estimated in both models.

Results

At baseline, women were on average 21 years old, 80.6% were white and 65.6% were in a stable relationship. The average age of sexual debut of these women was 17 years old, and 37 (2.2%) mentioned having been pregnant in the past, with 24 of these pregnancies ending in any abortion. Most women were using contraception at survey baseline (94.8%) with 67.2% using the pill, 22.5% using condoms, 5.1% using less effective methods (with-drawal/fertility-awareness/spermicide/diaphragms), and 5.2% reporting no use (data not shown in Table 1).

Among the 1679 women in the sample, 27.63% (n = 464) discontinued their method for method-related reasons and 18.2% (n = 305) abandoned or switched to a less effective method (Fig. 1). Pills more likely to be used by students who were older, white, had more years of sexual experience, who only had one lifetime partner, and who were in steady relationships. Conversely, students who are younger, black, in casual relationships, have more than one lifetime sexual partner, and more recently began having sex were more likely to rely on condoms at baseline (Table 1).

Among a total of 12,378 months of pill use, 2.7% of monthly-events represented discontinuation for methodrelated reasons. In addition, 3.3% of monthly-events represented switching from the pill to a less effective method or to no method. Among a total of 3581 months of condom use, 9.5% of monthly-events represented discontinuation for method-related reasons. In addition, 5.2% of monthly-events represented switching from the condom to a less effective method or to no method. Most women discontinued the pill because of side effects (66%), while condom users discontinued to switch to more effective methods (62%) or because their partners disliked the method (18%) (data not shown in Table).

Bivariate analyses of factors related to monthly discontinuation are reported in Table 2. Results indicate factors related to method discontinuation and abandonment/or switch to less effective contraception were mostly similar for both methods. The exception was time since first sex, which it was significantly related to discontinuation but not to method abandonment. Women who had been pregnant before were more likely to discontinue condoms but not to abandon condoms for less effective or no methods. In addition, condom users who were younger were more likely to abandon contraceptive use or switch to a less effective contraception (e.g., withdrawal, fertility-awareness, or spermicide). Results from the GEE **Table 1** Sociodemographic and sexual and reproductive health characteristics by type of contraceptive method used at baseline.São Paulo, Brazil – 2015

Variables	Total	Oral pill (N = 1190)	Condom (N = 398)	p*
	Ν	%	%	
Sociodemographic characteristics				
Age (years)				< 0.001
18–19	379	66.8	33.2	
20–24	1209	77.5	22.5	
Race/ethnicity				0.018
White (caucasian)	1280	76.4	23.6	
Black	217	67.7	32.3	
Other race/ethnicity**	91	71.4	28.6	
Religion				0.025
No religion	728	73.5	26.5	
Roman Catholic	501	77.1	22.9	
Kardecist Spiritualism Doctrine	196	81.1	18.9	
Evangelical	97	69.1	30.9	
Other religion***	66	65.2	34.8	
Type of relationship				0.001
Steady	1044	77.5	22.5	
Casual relationship or None	544	70.0	30.0	
Level of status				0.099
A/B	1259	75.9	24.1	
C – D/E	329	71.4	28.6	
Educational background				
Field of study				0.089
Human Sciences	643	72.3	27.7	
Health Sciences	588	75.7	24.3	
Exact Sciences	357	78.4	21.6	
Period of study				0.105
Full-time	814	76.7	23.3	
Part-time (morning, afternoon and evening)	774	73.1	26.9	
Campus				< 0.001
São Paulo	1039	71.8	28.2	
Out of São Paulo	549	19.1	80.9	
Sexual, reproductive and contraceptive behavior				
Time since first sexual intercourse				< 0.001
\leq 1 years	312	66.0	34.0	
2–3 years	561	74.0	26.0	
\geq 4 years	715	79.6	20.4	
Number of sexual partners in lifetime				0.005
1 partner	576	79.2	20.8	
2–3 partners	455	70.3	29.7	
≥4 partners	557	74.3	25.7	
Previous pregnancy				0.321
No	1554	75.1	24.9	

	1588	74.9	25.1	
Tatal	1500	74.0	25.1	
Yes	34	67.7	32.3	
	Ν	%	%	
Variables	Total	Oral pill ($N = 1190$)	Condom (<i>N</i> = 398)	p*

Table 1 Sociodemographic and sexual and reproductive health characteristics by type of contraceptive method used at baseline. São Paulo. Brazil – 2015 (*Continued*)

*Pearson's chi-square test

**Race/ ethnicity include: asian origin and indigenous people

***Other religions include: Afro-Brazilian, Buddhism, Jewish, Muslim, Mormon, and Islam

multivariate models presented in Table 3, show that partnership type was the strongest factor related to monthly discontinuation. Pill and condom users in casual relationships were at increased odds of monthly discontinuation. Furthermore, a prior pregnancy increased the odds of pill and condom discontinuation. Other determinants were method specific. For instance, pill users of lower socioeconomic status or who reported multiple lifetime partners (more than one sexual partner in lifetime) had higher odds to discontinue or abandon/switch to less effective method. In contrast, women decreased odds to discontinue the condom as they became more sexually experienced (as measured by years since first sexual intercourse).

Discussion

This is the first study to provide estimates of frequency of discontinuation among undergraduate women in Brazil, using a retrospective calendar data. It is original in providing estimates of switching and abandonment of contraceptive method. We focus our analysis on monthly odds of method discontinuation among women who have used contraception within 12-months period and who remain at risk of unintended pregnancy in order to identify sociodemographic and sexual health characteristics that are related to method discontinuation, which it can contribute increased risk of unintended pregnancy.

Our data shows that discontinuation is a relatively frequent event among undergraduate women in Brazil: almost three out of 10 women had at least one episode of discontinuation for method-related reasons, while two out of 10 women abandoned their method or switched to less effective method. Although our sample is different from 1996 DHS [18] (the sampling strategy involved geographical strata and it was representative of the country as a whole), and therefore comparison is limited, we found similar patterns of discontinuation. Nowadays, more women use contraception than do in 1996; however, this may indicate that discontinuation has changed little over the years in Brazil, despite upgraded health



Table 2 Proportion of all months	of contraceptive discontinuation	n according to the type	e of discontinuation, k	by type of
contraceptive method. São Paulo,	Brazil – 2015			

Variables		Oral Condom Pill		Discontinuation for method-related reasons			Abandonment or switching to less effective method			
			Oral	l Pill	Condom		Oral Pill		Condom	
	Ν	Ν	%	p*	%	p*	%	p*	%	p*
Sociodemographic characteristics										
Age (years)				0.301		0.874		0.489		0.032
18–19	2664	1053	2.4		9.4		3.1		6.5	
20-24	9714	2528	2.7		9.6		3.4		4.7	
Race/ethnicity				0.137		0.165		0.501		0.661
White (caucasian)	10,186	2719	2.6		9.9		3.2		5.4	
Black	1515	660	3.3		7.6		3.8		4.5	
Other race/ethnicity**	677	202	1.9		10.4		3.4		4.9	
Religion				0.029		0.348		0.029		0.817
No religion	5554	1776	2.7		8.9		3.5		5.1	
Roman Catholic	102	968	2.2		9.9		2.8		5.0	
Kardecist Spiritualism Doctrine	55	358	2.7		10.3		3.7		5.3	
Evangelical	23	260	2.9		8.1		3.4		5.8	
Other religion***	180	219	4.7		12.8		5.4		6.8	
Type of relationship at the time of discontinuation				< 0.001		0.054		< 0.001		0.052
Steady	8733	2179	2.2		8.8		2.8		5.4	
Casual relationship	3645	1402	3.7		10.7		4.6		4.9	
Level of status				0.020		0.473		0.032		0.315
A/B	9946	2705	2.5		9.7		3.2		5.4	
C – D/E	2432	876	3.3		8.9		4.0		4.6	
Educational Background										
Field of study				0.045		0.510		0.234		0.563
Human Sciences	4822	1591	3.1		10.1		3.6		5.3	
Health Sciences	4662	1310	2.3		8.8		3.0		5.5	
Exact Sciences	2894	680	2.5		9.4		3.3		4.4	
Period of study				0.002		0.830		0.037		0.907
Full-time	6514	1700	2.2		9.4		3.0		5.2	
Part-time	5864	1881	3.1		9.6		3.7		5.3	
Campus				0.007		0.134		0.043		0.340
São Paulo	7761	2672	2.9		9.1		3.4		5.0	
Out of São Paulo	4617	909	2.1		10.8		3.2		5.8	
Time since first sexual intercourse				0.046		0.050		0.314		0.927
\leq 1 years	2099	685	2.1		12.0		2.8		5.4	
2–3 years	4372	1368	2.4		8.8		3.3		5.0	
\geq 4 years	5907	1528	3.0		9.1		3.5		5.3	
Number of sexual partners in lifetime				< 0.001		0.666		< 0.001		0.612
1 partner	4783	1011	1.6		9.2		2.0		4.6	
2–3 partners	3355	1079	2.7		10.2		3.5		5.6	
≥4 partners	4240	1491	3.8		9.3		4.7		5.4	
Previous pregnancy				< 0.001		0.053		0.001		0.320
No	365	3482	2.5		9.4		3.2		5.3	

contraceptive method. São Paulo,	Brazil – 2015 (Continu	ied)					
Variables	Oral Pill	Condom	Discontinu method-re	ation for lated reasons	Abandonment or switching to less effective method		
			Oral Pill	Condom	Oral Pill	Condom	
	N	Ν	% p*	% p*	% p*	% p*	
Yes	18	99	7.7	15.1	7.2	3.0	
Total	12,378	3581	2.7	9.5	3.3	5.2	

Table 2 Proportion of all months of contraceptive discontinuation according to the type of discontinuation, by type of contraceptive method. São Paulo, Brazil – 2015 (*Continued*)

*Pearson's chi-square test

**Other race/ ethnicity include: asian origin and indigenous people

***Other religions include: Afro-Brazilian, Buddhism, Jewish, Muslim, Mormon, and Islam

services and improvement in some women's health indicators [29].

These relatively high rates of discontinuation are likely related to the method mix that are shown to have higher discontinuation rates [26]. Lack of professional training and limited contraceptive options [30, 31] reflect the difficulty for women to maintain contraceptive use or to adopt a new method after stopping their previous method. In addition, theses factors also contribute to a skewed method mix favoring short acting user dependent methods, especially among youth in Brazil who are at the peak of their fertility [30]. However, we emphasize that the desire for a more effective method represents a positive outcome of condom discontinuation for method-related reasons.

As previously reported in studies conducted in the U.S. or in France [28, 32, 33], countries with moderate to high contraceptive prevalence such as in Brazil, condom users were more likely to discontinue their method than pill users, but the reverse was true in case of method abandonment/switching to a less effective method. These results indicate the need to consider several indicators of discontinuation in order to better capture transitions that put women at risk of pregnancy from transitions that potentially improve pregnancy prevention. Regardless of the method used, partnership was the most significant variable of method discontinuation and abandonment. Other studies have pointed to the strong connection between partnership and contraceptive use dynamics. Among couples, condoms are mainly used during casual relationships or in the beginning of a relationship to prevent against Sexually Transmitted Infections (STIs) and are replaced by more effective contraception in more stable relationships [25, 34]. As relationships become more stable, many couples switch to hormonal methods, and the main reason to switch is the "supposed mutual trust". It denotes that they become less concerned about the prevention of the STIs, which it still remains an important public health problem in Brazil [35].

Our analysis also indicates that discontinuation is higher in casual relationships. These findings align with work conducted by Sassler and colleagues [36], which indicates that couples that believe they have a future together are more consistent users. Other research in the US, however, notes that women living with their sexual partner are more likely to have gaps in contraception than women who do not live with their sexual partner [37]. Study population and context may account for part of the differences observed between studies, but all studies draw attention to partner effects [12].

We also found that as the number of sexual partners increased, the odds of pill discontinuation increased. This result may relate to our prior discussion about the need for STI prevention with new partners, which may lead some women to switch from pills to condoms when they start a new relationship [34]. In contrast, condom users at the start of their sexual lives (less than 1 year of sexual experience) were more likely to discontinue, a finding that corroborates the results of a previous study in Honduras [38].

As expected, discontinuation of condoms was mainly related to criticism of the partner and wanting a more effective method. Regarding to negotiate condom use with sexual partners, studies relate many social and cultural barriers. The perception of pleasure, normative beliefs, lack of precise information, gender disadvantage, stereotypical gender roles and communication barriers have been shown to adversely affect lifetime condom use, which contribute to discontinuation of this method [39, 40]. Consequently, this scenario stands as a barrier to the prevention of STIs. On the other hand, discontinuation of hormonal contraception observed was related to side effects, such as also noticed in the literature [35, 41]. A study conducted with hormonal contraceptive users in nine countries reported that switching from oral pills because of side effects was common among Brazilian women. This suggests that women's dissatisfaction with their method has relevant consequences for contraceptive use [42].

Women's satisfaction with the contraceptive method is determined according to the characteristics of the methods itself [43]. In this context, studies suggest that

Variables	Discontinuation for r	method-related reasons	Abandonment or switching to less effective method		
	Oral Pill	Condom	Oral Pill	Condom	
	OR _{adjust} (CI)*	OR _{adjust} (CI)*	OR _{adjust} (CI)*	OR _{adjust} (CI)*	
Sociodemographic characteristics					
Age (years)					
18–19	1.0	1.0	1.0	1.0	
20–24	1.0 (0.8–1.4)	1.2 (0.9–1.7)	1.1 (0.8–1.4)	0.7 (0.5–0.9)**	
Race/ethnicity					
White (caucasian)	1.0	1.0	1.0	1.0	
Black	1.1 (0.8–1.6)	0.7 (0.5–1.1)	1.1 (0.8–1.5)	0.9 (0.6–1.3)	
Other race/ethnicity****	0.7 (0.4–1.3)	1.0 (0.6–1.8)	1.0 (0.6–1.5)	0.9 (0.5–1.8)	
Religion					
No religion	1.0	1.0	1.0	1.0	
Roman Catholic	0.9 (0.7–1.3)	1.1 (0.8–1.6)	0.9 (0.7–1.2)	0.9 (0.7–1.3)	
Kardecist Spiritualism Doctrine	1.1 (0.8–1.6)	1.1 (0.7–1.7)	1.2 (0.9–1.6)	1.1 (0.6–1.7)	
Evangelical	1.3 (0.8–2.1)	1.0 (0.6–1.7)	1.2 (0.8–1.9)	1.3 (0.7–2.2)	
Other religion*****	1.6 (0.9–2.7)	1.5 (0.9–2.6)	1.5 (0.9–2.4)	1.4 (0.8–2.5)	
Type of relationship at the time of	discontinuation				
Steady	1.0	1.0	1.0	1.0	
Casual relationship	1.4 (1.1–1.8)**	1.3 (1.0–1.7)**	1.4 (1.1–1.7)**	0.8 (0.6–1.1)	
Level of income					
A/B	1.0	1.0	1.0	1.0	
C – D/E	1.4 (1.0–1.8)**	0.9 (0.7–1.3)	1.3 (1.0–1.7)**	0.9 (0.7–1.3)	
Educational background					
Field of study					
Human Sciences	1.0	1.0	1.0	1.0	
Health Sciences	0.9 (0.7–1.3)	0.8 (0.5–1.2)	1.0 (0.7–1.3)	1.1 (0.7–1.6)	
Exact Sciences	1.1 (0.7–1.5)	0.8 (0.6–1.3)	1.0 (0.8–1.4)	0.8 (0.5–1.2)	
Period of study					
Full-time	1.0	1.00	1.0	1.0	
Part-time	1.2 (0.9–1.7)	0.9 (0.6–1.3)	1.2 (0.9–1.5)	1.0 (0.7–1.5)	
Campus					
São Paulo	1.0	1.0	1.0	1.0	
Out of São Paulo	0.8 (0.6–1.0)**	1.3 (0.9–1.7)	1.0 (0.8–1.2)	1.3 (0.9–1.9)	
Sexual and reproductive behavior					
Time since first sexual intercourse					
≤ 1 year	1.0	1.0	1.0	1.0	
2–3 years	0.9 (0.6–1.3)	0.5 (0.4–0.9)**	0.9 (0.7–1.3)	0.9 (0.6–1.4)	
≥ 4 years	0.9 (0.6–1.3)	0.6 (0.4–0.9)**	0.7 (0.5–1.1)	1.1 (0.7–1.7)	
Number of sexual partners in lifetin	ne				
1 partner	1.0	1.0	1.0	1.0	
2–3 partners	1.7 (1.2–2.4)**	1.1 (0.7–1.6)	1.8 (1.4–2.5)**	1.3 (0.9–2.0)	
≥4 partners	1.2 (1.6–3.2)**	0.9 (0.6–1.3)	2.4 (1.8–3.3)**	1.3 (0.9–2.1)	
Previous pregnancy					

Table 3 Correlates of contraceptive discontinuation according to the type of discontinuation, by type of contraceptive method. São Paulo, Brazil – 2015

Variables	Discontinuation for method-related reasons		Abandonment or switching to less effective method			
	Oral Pill	Condom	Oral Pill	Condom		
	OR _{adjust} (CI)*	OR _{adjust} (CI)*	OR _{adjust} (CI)*	OR _{adjust} (CI)*		
No	1.0	1.0	1.0	1.0		
Yes	2.7 (1.6–3.7)**	1.8 (0.9–2.6)**	2.1 (1.3–3.6)**	0.6 (0.2–1.8)		

Table 3 Correlates of contraceptive discontinuation according to the type of discontinuation, by type of contraceptive method. São Paulo, Brazil – 2015 (*Continued*)

*OR: odds ratios estimated using Generalized Estimating Equations with exchangeable matrix and logit link function assumed. CI: Confidence Interval ** p ≤ 0.050

***Other race/ ethnicity include: asian origin and indigenous people

****Religions include: Roman Catholic, Kardecist Spiritualism Doctrine, Evangelical, Afro-Brazilian, Buddhism, Jewish, Muslim, Mormon, and Islam

reproductive planning actions should prioritize the provision of LARCs, not only because they are associated with high satisfaction, but also because they are highly effective, avoid the need for frequent visits for resupply, they are highly cost effective, and allow a rapid return to fertility after their removal; although they offer no protection against the STIs [17, 44]. Additionally, the provision of LARCs in the Brazilian public health system could modify the Brazilian contraceptive mix, with positive effects in decreasing the occurrence of unintended pregnancies and induced abortions [43].

Despite our selection of a relatively privileged segment of the Brazilian population, we found increased odds of pill disruption among students with lower socioeconomic status. Lower income women are less likely to use contraception compared to women who are better off financially [45]. In Brazil, women have free access to contraception through the public health system, yet delays in appointment may contribute to contraceptive gaps or switches to non-prescription contraception. Although not legally authorized, many women obtain their pills from pharmacists with no prescription, but out of pocket cost may not be an option for the poorest students. In this context, we pointed out that the educational level was not able to void the effect of the socioeconomic status, and this outcome is important in a country as unequal as Brazil. Further investigation of the differential reasons for pill discontinuation by socioeconomic status would shed more light on these findings and inform policy decisions regarding improvement of the quality of family planning services in primary health care.

Another important finding was the association between age and method abandonment. Consistent with our results, study that use data of the National Survey of Family Growth observed that condom users who were younger were almost 42% more likely to discontinue the method compared to older women [28]. Additionally, the authors observed that condom users, who never became pregnant, were 27% less likely to discontinue for method related-reasons when compared to women with children, which is also compatible with our results about the parity and contraceptive discontinuation [28]. Finally, educational background was associated to discontinuation. Pill users who were studying out of the city of São Paulo were 20% less likely to discontinue. This finding was surprising, since São Paulo has more 24 h pharmacies available and primary health care facilities. This is the first study that addressed educational characteristics in the context of discontinuation and more qualitative research is needed to understand the associations observed.

The high frequency of discontinuation observed in our population calls for greater attention to the difficulties Brazilian women face when using short acting methods. Implants and hormonal intrauterine device are only available in Brazil at private clinics and cooper intrauterine device are rarely offered in the public health service [1], leaving women with suboptimal options to prevent a pregnancy. This is particularly concerning in a country that does not legally provide access to abortion when contraception fails.

Contraceptive discontinuation is a particularly significant issue for adolescents and young women, since they tend to have more limited access, and face more barriers than older women in family planning programs. Furthermore, they have more unpredictable and irregular sexual activity; therefore, high quality family planning services are extremely necessary. Adequate, accurate and detailed information on the importance of consistent contraceptive use should be emphasized in order to minimize contraceptive discontinuation, whether during medical consultations or students discussion meetings around the units.

This study has a number of strength and limitations. While the use of probability sampling and a relatively high response rate for an online survey increased the internal validity of the study, our sample is not representative of all Brazilian undergraduate students, such as those at private universities and less educated women. In spite of several incentive programs that favor the inclusion of students from less privileged socioeconomic status at University of São Paulo, studies carried out in some campuses at University of São Paulo observed that the majority of young students is white, belongs to high/

medium socioeconomic status, studied in private schools, and the parents of these students have a college degree [46, 47]. Thus, the profile of the young students who participated in this study differs from the other Brazilian youths; however, these are valuable findings and clarify what is happening with this relatively privileged population.

The frequency of discontinuation in this segment of the population calls for population based research to assess the frequency of the issue across the socioeconomic spectrum in Brazil. In this study, we used calendar data to assess monthly use of contraception, allowing for a refined analysis of discontinuation. Retrospective data however, may suffer from recall bias. Additionally, we analyzed only 12-months, nevertheless the recall bias about sexual and contraceptive behaviors may have been minimized by the fact that the study population was composed of young women who initiated the sexual intercourse recently.

Conclusion

Contraceptive discontinuation is a frequent event among undergraduate students in Brazil, who mostly rely on short-acting methods for pregnancy prevention. Contraceptive discontinuation does differ by type of contraceptive method, condom users were more likely to discontinued their method for method-related reasons, whereas oral pill users were more likely to abandonment or switching to a less effective method. The partnership context was the strongest predictor of discontinuation, emphasizing the importance of relational factors contributing pregnancy risk exposure. Relationship type contributes to method choice, and method discontinuation. The limited contraceptive options in the public health system in Brazil probably reflect the difficulty for women to maintain contraceptive use, and actions that promote the provision of long-term methods should be taken into account in order to minimize the impacts of conceptive discontinuation. We believe that the public health impact of contraception will not be concretized until all women who desire to prevent pregnancies are using their method of choice continuously and effectively.

Abbreviations

DHS: Demographic and Health Survey; GEE: Generalized Estimating Equation; QIC: Independence Model Criterion; STIs: Sexually Transmitted Infections; U.S.: United States

Acknowledgements

This work was supported by the Brazilian National Counsel of Technological and Scientific Development/ CNPq (202219/2015-1) and the Brazilian Coordination for the Improvement of Higher Education Personnel/CAPES.

Authors' contributions

CBNC and ALVB conceived and conducted the study, made substantial contributions to interpretation of data and wrote the manuscript. CBNC and

CM took the responsibility for the accuracy of data analyses, and interpretation of data. CM and OAS made substantial contributions to interpretation of data, and co-wrote the manuscript. All authors read and approved the final manuscript.

Funding

This work was supported by the Conselho Nacional de Desenvolvimento Científico e Tecnológico/CNPq [grant number 202219/2015–1]. This study was financed in part by the Coordenação de Aperfeiçoamento de Pessoal de Nível Superior - Brasil (CAPES) - Finance Code 001.

Availability of data and materials

The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request.

Ethics approval and consent to participate

Data collection was performed after approval of this project by the Research Ethics Committee of the University of São Paulo School of Nursing (CAAE Number: 39142514.4.0000.5392).

Consent for publication

Not applicable.

Competing interests

The authors declare no competing interests.

Author details

¹ Direct-Entry Midwifery Program, School of Arts, Sciences and Humanities of the University of São Paulo, Av. Arlindo Bettio, 1000 – Ermelino Matarazzo, São Paulo, São Paulo 03828-000, Brazil. ²Department of Population, Family and Reproductive Health, Johns Hopkins Bloomberg School of Public Health, Baltimore, MD, USA. ³Gender, sexual and reproductive health, CESP Centre for research in Epidemiology and Population Health, U1018, Inserm, F-94807 Le Kremlin-Bicêtre, France. ⁴Department of Public Health Nursing, University of São Paulo School of Nursing, São Paulo, Brazil.

Received: 25 May 2018 Accepted: 28 July 2019 Published online: 29 August 2019

References

- Ministério da Saúde, Centro Brasileiro de análise e Planejamento. PNDS 2006: Pesquisa Nacional de Demografia e Saúde da Criança e da Mulher; 2009. http://bvsms.saude.gov.br/bvs/publicacoes/pnds_crianca_mulher.pdf. Acessed 15 Nov 2018.
- Sociedade Civil Bem-Estar Familiar no Brasil BEMFAM, Pesquisas Demograficas e de Saúde, and Institute for Resource Development/Macro Systems. 1987. Pesquisa Nacional Sobre Saúde Materno-Infantil e Planejamento Familiar Brasil 1986. Rio de Janeiro, Brasil: BEMFAM, Pesquisas Demograficas e de Saúde, and the Institute for Resource Development/Macro Systems.
- Viellas EF, Domingues RMSM, Dias MAB, Gama SGN, Theme Filha MM, Costa JV, Bastos MH, Leal MC. Prenatal care in Brazil. Cad Saúde Pública. 2014; 30(Suppl1):S85–S100. https://doi.org/10.1590/0102-311X00126013.
- Moreau C, Bohet A, Trussell J, Bajos N, FECOND Group. Estimates of unintended pregnancy rates over the last decade in France as a function of contraceptive behaviors. Contraception. 2014;89(4):314–21. https://doi.org/1 0.1016/j.contraception.2013.11.004.
- Wellings K, Brima N, Sadler K, Copas AJ, McDaid L, Mercer CH, et al. Stopping and switching contraceptive methods: findings from Contessa, a prospective longitudinal study of women of reproductive age in England. Contraception. 2015;91(1):57–66. https://doi.org/10.1016/j.contraception.2014.09.008.
- Finer LB, Zolna MR. Declines in unintended pregnancy in the United States, 2008-2011. N Engl J Med. 2016;374(9):843–52. https://doi.org/10.1 056/NEJMsa1506575.
- Creanga AA, Acharya R, Ahmed S, Tsui AO. Contraceptive discontinuation and failure and subsequent abortion in Romania: 1994-99. Stud Fam Plan. 2007;38(1):23–34.
- Kost K, Singh S, Vaughan B, Trussell J, Bankole A. Estimates of contraceptive failure from the 2002 National Survey of family growth. Contraception. 2008;77(1):10–21. https://doi.org/10.1016/j.contraception.2 007.09.013.

- Bajos N, Lamarche Vadel A, Gilbert F, Ferrand M, Moreau C. Cocon group. Contraception at the time of abortion: high-risk period or highrisk women? Hum Reprod. 2006;21:2862–7.
- 10. Leite IC, Gupta N. Assessing regional differences in contraceptive discontinuation, failures and switching in Brazil. Reprod Health. 2007;4:6.
- Bradley SEK, Schwandt HM, Khan S. Levels, Trends, and reasons for contraceptive discontinuation. DHS: demographic and Health Research division: report. United States; 2009. https://dhsprogram.com/pubs/pdf/ AS20/AS20.pdf. Acessed 15 Nov 2018
- Barden-O'Fallon J, Speizer IS, Cálix J, Rodriguez F. An analysis of contraceptive discontinuation among female, reversible method users in urban Honduras. Stud Fam Plan. 2011;42(1):11–20 https://www.ncbi.nlm. nih.gov/pmc/articles/PMC3883041/pdf/nihms538066.pdf. Acessed 15 Nov 2018.
- Moreau C, Hall K, Trussell J, Barber J. Effect of prospectively measured pregnancy intentions on the consistency of contraceptive use among young women in Michigan. Hum Reprod. 2013;28(3):642–50. https://doi. org/10.1093/humrep/des421.doi:10.1093/humrep/des421.
- Manlove J, Welti K, Barry M, Peterson K, Schelar E, Wildsmith E. Relationship characteristics and contraceptive use among young adults. Perspect Sex Reprod Health. 2011;43:119–29. https://doi.org/1 0.1363/4311911.
- Manning WD, Flanigan CM, Giordano PC, Longmore MA. Relationship dynamics and consistency of condom use among adolescents. Perspect Sex Reprod Health. 2009;41(3):181–90. https://doi.org/10.1363/4118109.
- Instituto Brasileiro de Geografia e Estatística. Pesquisa Nacional de Saúde: 2013. Ciclos de vida: Brasil e grandes regiões, vol. 3. 1st ed. Rio de Janeiro: IBGE; 2015. https://biblioteca.ibge.gov.br/visualizacao/livros/liv94522.pdf. Acessed 15 Nov 2018
- Trussell J. Contraceptive failure in the United States. Contraception. 2011; 83(5):397–404. https://doi.org/10.1016/j.contraception.2011.01.021.
- Ali MM, Cleland J, Shah IH. Causes and consequences of contraceptive discontinuation: evidence from 60 demographic and health surveys. Geneva: World Health Organization; 2012. http://apps.who.int/iris/bitstream/handle/1 0665/75429/9789241504058_eng.pdf?sequence=1. Acessed 15 Nov 2018
- Sant'Anna MJC, Carvalho KAM, Passarelli MLB, Coates V. Comportamento sexual entre jovens universitários. Adolesc Saude. 2008;5(2):52–6.
- Borges ALV, Fujimori E, Hoga LAK, Contim M. Práticas contraceptivas entre jovens universitários: uso da anticoncepção de emergência. Cad Saude Publica. 2010;26(4):816–26. https://doi.org/10.1590/S0102-311X20100004 00023.
- Coetzee MH, Ngunyulu RN. Assessing the use of contraceptives by female undergraduate students in a selected higher educational institution in Gauteng. Curationis. 2015;38(2):1535. https://doi.org/10.4102/curationis.v3 8i2.1535.
- Bryant KD. Contraceptive use and attitudes among female college students. ABNF J. 2009;20(1):12–6.
- Trieu SL, Shenoy DP, Bratton S, Marshak HH. Provision of emergency contraception at student health centers in California community colleges. Womens Health Issues. 2011;21(6):431–7. https://doi.org/10.1016/j.whi.2011.04.011.
- Associação Brasileira de Empresas de Pesquisa. Critério de classificação econômica Brasil. Dados com base no levantamento socioeconômico. 1st ed. São Paulo: ABEP; 2015. http://www.abep.org/criterio-brasil. Acessed 15 Nov 2018
- Manlove J, Ryan S, Franzetta K. Contraceptive use patterns across teens' sexual relationships: the role of relationships, partners, and sexual histories. Demography. 2007;44(3):603–21.
- Curtis KM, Jatlaoui TC, Tepper NK, Zapata LB, Horton LG, Jamieson DJ, et al. U.S. selected practice recommendations for contraceptive use. MMWR Recomm Rep. 2016;65(4):1–66 https://www.cdc.gov/mmwr/volumes/65/rr/ rr6504a1.htm. Acessed 15 Nov 2018.
- Trussell J. Contraceptive efficacy. In: Hatcher RA, Trussell J, Nelson AL, Cates W, Stewart FH, Kowal D, editors. Contraceptive technology: nineteenth revised edition. New York: Ardent Media; 2007. p. 747–826.
- Vaughan B, Trussell J, Kost K, Singh S, Jones R. Discontinuation and resumption of contraceptive use: results from the 2002 National Survey of family growth. Contraception. 2008;78(4):271–83. https://doi.org/10.1 016/j.contraception.2008.05.007.
- Victora CG, Aquino EM, Do Carmo Leal M, Monteiro CA, Barros FC, Szwarcwald CL. Maternal and child health in Brazil: Progress and challenges. Lancet. 2011; 377(9780):1863–76. https://doi.org/10.1016/S0140-6736(11)60138-4.

- Heilborn ML, Portella AP, Brandão ER, Cabral C d S. Assistência em contracepção e planejamento reprodutivo na perspectiva de usuárias de três unidades do Sistema Único de Saúde no Estado do Rio de Janeiro, Brasil. Cad Saude Publica. 2009;25(Suppl 2):s269–78. https://doi. org/10.1590/S0102-311X2009001400009.
- Vieira EM, Souza L. A satisfação com o serviço de esterilização cirúrgica entre os usuários do Sistema Único de Saúde em um município paulista. Rev Bras Epidemiol. 2011;14(4):556–64. https://doi.org/10.1590/ S1415-790X2011000400003.
- Mansour D, Korver T, Marintcheva-Petrova M, Fraser IS. The effects of Implanon on menstrual bleeding patterns. Eur J Contracept Reprod Health Care. 2008;13(Suppl 1):13–28. https://doi.org/10.1080/136251 80801959931.
- Moreau C, Bouyer J, Bajos N, Rodríguez G, Trussell J. Frequency of discontinuation of contraceptive use: results from a French populationbased cohort. Hum Reprod. 2009;24(6):1387–92. https://doi.org/10.1093/ humrep/dep027.
- Lantos H, Bajos N, Moreau C. Determinants and correlates of preventive behaviors at first sex with a first partner and second partner: analysis of the FECOND study. J Adolesc Health. 2016;58(6):644–51. https://doi. org/10.1016/j.jadohealth.2016.03.006.
- Mehra D, Östergren PO, Ekman B, Agardh A. Inconsistent condom use among Ugandan university students from a gender perspective: a cross-sectional study. Glob Health Action. 2014;10(7):229–42. https://doi. org/10.3402/gha.v7.22942.
- Sassler S, Miller A, Favinger SM. Planned Parenthood?: fertility intentions and experiences among cohabiting couples. J Fam Issues. 2008;30(2): 206–32. https://doi.org/10.1177/0192513X08324114.
- Frost JJ, Singh S, Finer LB. U.S. women's one-year contraceptive use patterns, 2004. Perspect Sex Reprod Health. 2007;39(1):48–55. https://doi. org/10.1363/3904807.
- Barden-O'Fallon J, Speizer I. What differentiates method stoppers from switchers? Contraceptive discontinuation and switching among Honduran women. Int Perspect Sex Reprod Health. 2011;37(1):16–23. https://doi.org/10.1363/3701611.
- Sarkar NN. Barriers to condom use. Eu r J Contracept Reprod Health Care. 2008;13(2):114–22. https://doi.org/10.1080/13625180802011302.
- Pinto VM, Basso CR, Barros CRS, Gutierrez EB. Factors associated with sexually transmitted infections: a population based survey in the city of São Paulo, Brazil. Ciênc saúde coletiva. 2018;23(7):2423–32. https://doi. org/10.1590/1413-81232018237.20602016.
- Inoue K, Barratt A, Richters J. Does research into contraceptive method discontinuation address women's own reasons? A critical review. J Fam Plann Reprod Health Care. 2015;41(4):292–9. https://doi.org/10.1136/ jfprhc-2014-100976.
- Hooper DJ. Attitudes, awareness, compliance and preferences among hormonal contraception users: a global, cross-sectional, selfadministered, online survey. Clin Drug Investig. 2010;30(11):749–63. https://doi.org/10.2165/11538900-00000000-00000.
- Borges ALV, Santos OA, Araújo KS, Gonçalves RFS, Rosa PLFS, Nascimento NC. Satisfaction with the use of contraceptive methods among women from primary health care services in the city of São Paulo, Brazil. Rev Bras Saude Mater Infant. 2017;17(4):749–56. https://doi. org/10.1590/1806-93042017000400008.
- Wulifan JK, Brenner S, Jahn A, De Allegri M. A scoping review on determinants of unmet need for family planning among women of reproductive age in low and middle income countries. BMC Womens Health. 2016;16:2. https://doi. org/10.1186/s12905-015-0281-3.
- Jones J, Mosher W, Daniels K. Current contraceptive use in the United States, 2006-2010, and changes in patterns of use since 1995. Natl Health Stat Rep. 2012;60:1–25.
- Barreyro GB, Aureliano AF. Students of public universities of the state of São Paulo: new campuses, old inequalities? Educere et Educare. 2010;5(10):33–44.
- Ishii I, Krasilchik M, Leite RC. Diversidade de alunos: o caso da USP. R Educ Públ. 2014;23(54):681–700.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.